Author’s response to reviews

Title: MAMI: a birth cohort focused on maternal-infant microbiota during early life

Authors:

Izaskun Garcia-Mantrana (igama@iata.csic.es)
Cristina Alcántara (crisalba@iata.csic.es)
Marta Selma-Royo (mselma@iata.csic.es)
Alba Boix-Amorós (albaboix@iata.csic.es)
Majda Dzidic (majda.dzidic@gmail.com)
Jose Gimeno-Alcañiz (jgimeno@iata.csic.es)
Isabel Úbeda-Sansano (isabel.ubeda@gmail.com)
Ignacio Sorribes-Monrabal (isorribesm@gmail.com)
Ramon Escuriet (rescuriet@catsalut.cat)
Fernando Gil Raga (fgil@hospitalmanises.es)
Anna Parra Llorca (annaparrallorca@gmail.com)
Cecilia Martinez-Costa (ceciliam@comv.es)
Maria Carmen Collado (mcolam@iata.csic.es)

Version: 4 Date: 09 Apr 2019

Author’s response to reviews:

Dear Editor

we would like to highlight that figures included in this study protocol are from our own and where designed by the team members.

Please, let us know if something else is needed.

we would also profit this letter to inform you that we submit first time our work last july 2018, and it has been a quite long review process for a study protocol that does not contain data and results.